

# **Effect of oral polymeric diet enriched with TGF-beta 2 (Modulen) versus Budesonide on clinical response and mucosal healing in Crohn's patients**

## **Reviewer #1:**

Thank you for reviewing our paper.

Abstract line 18: The required change was done

Keywords: The required change was done

Methods: The required change was done

Results: Lines 17-20 provide a report of the results (number of patients who left the study and their reasons for withdrawal). These numbers are part of the results of any analysis. No discussion appears here.

Table 1: The required changes were done

Table 2: The required changes were done

Table 3: The required changes were done

Table 4: The required changes were done

Table 5: The required changes were done

Discussion: The required changes were done

## **Reviewer #2:**

Thank you for reviewing our paper.

Thank you for your profound analysis of the paper.

No changes were required.

## **Response to Editorial Office's comments:**

The title was changed as suggested.

The author list and institutions were changed as suggested.

Please notice that Baruch Ovadia and Eva Niv are first co-authors.

Please notice that the name of an additional author was added (Oren Gal). His name was missed by mistake at the first submission.

The "Author contribution" was added.

The "Key words" were changes as suggested.

The "Cor Tip" was added.

The "Audio Core Tip" was recorded.

The reference numbers were changed as suggested.

The PMID and DOI were added as suggested.

The tables were changed as suggested.

The abbreviations were changed as suggested.

The "Aim" was added to the abstract.

The "Conclusion" was added to the main text.

The title was changed as suggested.

The running title was added.

The supportive foundation was altered as suggested.

The corresponding author was changed as suggested.

The abstract was changed as suggested.